Author's response to reviews

Title: Actuarial survival of a large Canadian cohort of preterm infants

Authors:

Huw P Jones (joneshuw@doctors.org.uk)
Stella Karuri (skaruri@cw.bc.ca)
Catherine MG Cronin (ccronin@cc.umanitoba.ca)
Arne Ohlsson (aohlsson@mtsinaion.ca)
Abraham Peliowski (apeliows@cha.ab.ca)
Anne Synnes (asynnes@cw.bc.ca)
Shoo K Lee (shoollee@cha.ab.ca)
Canadian NEONATAL Network (shoollee@cha.ab.ca)

Version: 3 Date: 1 November 2005

Author's response to reviews:

October 31, 2005

The Editor, BMC Pediatrics

Dear Editor:

As requested, we have made the following changes. If there are any further changes required, please let us know.

Sincerely,
Shoo K Lee

Title & Abstract: These are identical in the manuscript and submission system

Author list: These are identical in the manuscript and submission system

Title: This has been changed to sentence case in the manuscript and submission system

Qualifications: These have been removed from the title page

Findings from this study: This section has been removed

Section headings: These have been changed to sentence case

Abstract: Structure has been changed as suggested

Introduction: This has been changed to Background

References: These have been changed to normal case numbers in square brackets

Subjects and methods: This has been changed to Methods

Members of the Canadian Neonatal Network: This has been moved to the Acknowledgements

References: All authors names have been included

Tables: These have been renamed in both the text and Tables.

Figures: The figures have been close cropped

Figures: The resolution of figure 9 has been improved
Additional file: This has been deleted